Author's response to reviews

Title: Evaluation of Phytochemical content, antimicrobial, cytotoxic and antitumor activities of extract from Rumex hastatus D.Don roots

Authors:

Rahmat Ali Khan (rahmatgul_81@yahoo.com)
Muhammad Rashid Khan (mrkhangau@yahoo.com)
Sumaria Sahreen (Sumariasahreen@gmail.com)
Taibi ben Hadda (Taibibenhada@gmail.com)

Version: 2 Date: 3 February 2015

Author's response to reviews: see over
Dear Editor

Refer to MS: 1404602610158606 title: Evaluation of Phytochemical content, antimicrobial, cytotoxic and antitumor activities of extract from Rumex hastatus D.Don roots

Find the revised manuscript for onward process please. All the suggestions are responded one by one.

1. Please revise your manuscript to include line and page numbers. Authors are asked to ensure that line numbering is included in the main text file of their manuscript at the time of submission to facilitate peer-review.

   Ans: Line and page numbers are included.

2. Acknowledgements: We strongly encourage you to include an Acknowledgements section between the Authors’ contributions section and Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source (s) of funding. Please also acknowledge anyone who contributed materials essential for the study.

   Ans: Acknowledgements section between the Authors’ contributions section and Reference list

3. Please ensure that the ORCiD identifier for author Rahmat Ali Khan is included in your Authors' Contributions section. As detailed on previous submissions, the inclusion of this information is necessary in order for us to be able to consider any submissions from Prof. Khan.

   Ans: ORCiD identifier for author Rahmat Ali Khan is included as (ORCID ID: 0000-0003-0453-2090)

4. Please provide details in your Methods section on who undertook the formal identification of the plant material used in your study. Please also confirm whether a voucher specimen of this material has been deposited in a publicly available herbarium, and include this information in your manuscript. A deposition number should be included, if available.

   Ans: Voucher specimen of this material has been deposited is included.